Author’s response to reviews

Title: Quality of life after the initiation of dialysis or maximal conservative management in elderly patients: A longitudinal analysis of the Geriatric assessment in OLder patients starting Dialysis (GOLD) Study

Authors:

Ismay van Loon (I.vanloon@dianet.nl)

Namiko Goto (n.a.goto@umcutrecht.nl)

Franciscus Boereboom (f.boereboom@dianet.nl)

Marianne Verhaar (m.c.verhaar@umcutrecht.nl)

Michiel Bots (m.l.bots@umcutrecht.nl)

Marije Hamaker (mhamaker@diakhuis.nl)

Version: 4 Date: 01 Feb 2019

Author’s response to reviews:

We've addressed the two minor revisions suggested by reviewer 2.